Please cite the Published Version

Rickett, Michelle Clare , Kingstone, Tom , Gupta, Veenu , Shiers, David, French, Paul , Lennox, Belinda, Crawford, Mike , Penington, Edward, Hedges, Anna, Ward, Jo, Williams, Ryan, Bateman, Paul and Chew-Graham, Carolyn A (2024) Collaboration across the primary/specialist interface in early intervention in psychosis services: a qualitative study. The British journal of general practice: the journal of the Royal College of General Practitioners. ISSN 0960-1643

DOI: https://doi.org/10.3399/bjgp.2023.0558

Publisher: Royal College of General Practitioners

Version: Published Version

Downloaded from: https://e-space.mmu.ac.uk/634982/

Usage rights: Creative Commons: Attribution 4.0

Additional Information: This is an open access article which first appeared in The British journal

of general practice: the journal of the Royal College of General Practitioners

Enquiries:

If you have questions about this document, contact openresearch@mmu.ac.uk. Please include the URL of the record in e-space. If you believe that your, or a third party's rights have been compromised through this document please see our Take Down policy (available from https://www.mmu.ac.uk/library/using-the-library/policies-and-guidelines)

Collaboration across the primary and specialist care interface in Early Intervention in Psychosis services: a qualitative study

Michelle Rickett, Tom Kingstone, Veenu Gupta, David Shiers, Paul French, Belinda Lennox, Mike Crawford, Ed Penington, Anna Hedges, Jo Ward, Ryan Williams, Paul A Bateman and Carolyn A Chew-Graham

Abstract

Background

People with new psychotic symptoms may be managed in an Early Intervention in Psychosis (EIP) service. They may be discharged back to primary care at the end of their time in an EIP service.

To explore the role of primary care in supporting people with psychosis in an EIP service.

Design and setting

Qualitative study, within a programme of work to explore the optimum duration of management in an EIP service in England.

Method

Semi-structured interviews were carried out with people in EIP services,

carers, GPs, and EIP practitioners between September 2022 and September 2023. Data collection continued until information power was achieved. Data were thematically analysed using principles of constant comparison.

Results

While most service users and carers described their experiences of EIP services as positive, there are issues around access to and discharge from the services. GPs reported difficulties in referring people into EIP services, having little contact with people who are supported by EIP services, and not being included in planning discharge from EIP services to primary care. Service users and carers described challenges at the point of

discharge from EIP services to primary care, associated with feelings of abandonment.

Conclusion

This study suggests that GPs should have a role in the support of people in EIP services (in particular, monitoring and managing physical health) and their carers. Inclusion of GPs in managing discharge from EIP services is vital. We suggest that a joint consultation with the service user, their carer (if they wish), along with the EIP care coordinator and GP would make this transition smoother.

Keywords

continuity of care; Early Intervention in Psychosis; mental illness; primary care; qualitative research; transitional care.

Introduction

People who develop new psychotic symptoms may present to primary care and be referred to, and accepted into, Early Intervention in Psychosis (EIP) services. These are multidisciplinary, community-based mental health teams offering treatment to people who experience a first episode of psychosis. EIP services are designed to intervene to reduce the duration of untreated psychosis, which has been associated with a worse prognosis.1 EIP service input has been shown to reduce negative outcomes such as coercive crisis management and hospitalisation under the Mental Health Act.² All service users should have a dedicated EIP care coordinator. Many are also supported

by informal caregivers, typically close relatives, including parents, children, partners, and siblings, who can experience lack of support and negative impacts on their own health and wellbeing.3

In the UK, EIP services are time limited, offering up to 3 years of treatment. Service users are then discharged either to a community mental health team (CMHT), which offers less intensive contact and interventions, or directly to primary care. Factors influencing discharge to a CMHT include enduring psychoses, referral to EIP from inpatient services, and longer time under EIP care.4 There is little guidance around planning and implementation of discharge from EIP to other services,5 particularly to primary care,6 even though most people are discharged to primary care.⁷ A previous study of transition from EIP services has suggested that better interagency collaboration and service user preparation for discharge, particularly the transition to primary care, is needed.8 The need for primary care to ensure that physical needs are met, even when service users are actively engaged with EIP services, has been emphasised previously.6

People with psychosis are at risk of developing physical health problems such as diabetes and cardiovascular disease, with the risk of early death.^{9,10} Physical health monitoring should be carried out at the start of treatment and at least annually, and this may happen in primary care or the EIP service.11 Some people

How this fits in

Early Intervention in Psychosis (EIP) service users may be referred from, and discharged back to, primary care. There is limited research on patients' and carers' experience of discharge to primary care from EIP services and little guidance around planning and implementation of discharge. This study explores experiences of EIP care and discharge from the perspectives of service users, carers, and healthcare professionals in EIP services and primary care. It explores the patient journey through EIP services, highlights the lost connection with primary care, and makes recommendations for more collaboration between primary and specialist care, particularly around physical health monitoring and management, which might improve patient experience and outcome.

may not have appropriate monitoring and management, while others may have duplicate assessments.¹²

The aim of this qualitative study was to understand and contextualise the experiences of EIP services and the role of primary care in supporting service users, from the perspectives of recipients and providers of care.

Method

This qualitative study was part of a larger National Institute for Health and Care Research-funded mixed methods programme (EXTEND: Personalised Care for Early Psychosis), which aims to examine the impact of duration of EIP care on patient outcomes.

Our qualitative methods were underpinned by interpretivism to support exploration of contexts, meanings, and interactions in relation to EIP services from multiple stakeholder perspectives. Semi-structured interviews were used to explore the views and experiences of EIP service users, carers, healthcare professionals from primary and specialist care, managers, and commissioners about EIP services.

Patient and public involvement (PPI) was integral to the study. Our two PPI co-investigators and service user and carer advisory group (EXTEND-InG) co-designed topic guides, public-facing documents, and recruitment strategies. They also contributed to the interpretation of data during analysis

meetings with the research team and via email feedback.

Recruitment of participants

We recruited EIP service users and carers at the time of, or shortly after, discharge from an EIP service (we excluded service users lacking capacity or expressing suicidal ideation), GPs, EIP practitioners and managers, and commissioners of mental health services.

Service users and carers were recruited using both purposive and convenience sampling. Potential participants were identified through mental health trusts (MHTs) across England who reported a mix of high and low duration of treatment. Other participants self-identified by responding to a study flyer shared via social media (X), mental health networks, support groups, and charities. The flyer included a QR code linking to the EXTEND website.

EIP practitioners and managers were identified through participating MHTs. GPs were identified through professional networks and snowballing; we targeted diversity in relation to locality, size, and type of practice.

An information sheet and 'consent to contact' form were used. On return of the 'consent to contact' form, individuals were contacted by the study researcher (by email or telephone), who checked their eligibility against inclusion criteria, and invited them to participate in an online (Microsoft Teams) or telephone interview, depending on the participant's preference. The researcher conducted screening calls before interviews with service users and carers to assess capacity and to check eligibility (including age, care under EIP service, and time of discharge). Service users were given the option to be interviewed with a carer.

Consent forms were completed electronically before or at the start of the interview. At the end of the interview, participants were asked if they wished to receive a plain English summary of the research findings and/or related publications. Service users and carers were offered a shopping voucher to recompense them for their time and GPs were reimbursed as per British Medical Association guidance.

Data generation

Semi-structured interviews were conducted by a female postdoctoral

M Rickett (ORCID: 0009-0001-2503-CA Chew-Graham (ORCID: 0000-0002-9722-9981), BSc, MD, FRCGP, professor of general University, Keele. **T Kingstone** (ORCID: 0000-Stafford. **V Gupta** (ORCID: 0000-0001-5082-University, Manchester. **D Shiers** (ORCID: 0000-0003-2531-5837), MRCP, FRCGP, carer; honorary research fellow, School o research consultant, Psychosis Research Unit, Greater Manchester Mental Health NHS Trust, Manchester. **P French** (ORCID: 0000-0003of Nursing and Public Health, Manches Metropolitan University, Manchester. **B Lennox** (ORCID: 0000-0002-4972-8786), DM, FRCPsych, professor of psychiatry; **E Penington** (ORCID: 0000-0001-7668-438X), BA, MSc, of Oxford, Oxford. M Crawford (ORCID: 0000 of mental health research; R Williams (ORCID: College London, London. A Hedges, MRCPysch, child and adolescent psychiatrist, Bolton CAMHS, Greater Manchester Mental Health NHS Foundation Trust, Manchester. **J Ward**, MSc, PA Bateman (ORCID: 0000-0003-0626-7809), MSc, MSci, PhD, statistical programmer, Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford.

CORRESPONDENCE Michelle Rickett

School of Medicine, David Weatherall Building,

Email: m.c.rickett1@keele.ac.uk

Submitted: 24 October 2023; Editor's response: 14 December 2023; final acceptance:

This is the full-length article (published online 20 Aug 2024) of an abridged version published in print. Cite this version as: **Br J Gen Pract 2024**; DOI: https://doi.org/10.3399/BJGP.2023.0558

researcher with qualitative research expertise. Interview participants were not given any information about the researcher other than that she was employed by Keele University.

A topic guide supported exploration of the experiences and views of EIP services, duration of care, decision making about discharge, and arrangements for ongoing care. Topic guides were developed with co-investigators and people with lived experience as EIP service users

or carers. The guides were modified iteratively alongside data generation and analysis. ^{13,14} Data collection continued until information power was achieved. Information power is an alternative concept to saturation in qualitative research and involves pragmatic judgements based on aims, specificity, theory, dialogue, and analysis. ¹⁵ Interviews were conducted between September 2022 and September 2023.

The interviews were audio-recorded, transcribed by a professional transcribing company, then checked and anonymised by the first author, who also compiled field notes after each interview. Analysis was performed using the qualitative research software NVivo (version 11). Transcripts were not returned to participants for comment and participants were not asked to give feedback on the findings. Data were thematically analysed using principles of constant comparison; this involved the researcher reading and re-reading transcripts, identifying and organising codes, and constructing themes.16 A framework approach was used to draw analysis from different participant groups together;¹⁷ the framework focused on the service user's journey through EIP services and the complexities around discharge. The first author led the analysis and met regularly with four other research team members, with a range of backgrounds and disciplines, to check data interpretation and agree themes. Members of EXTEND-InG also provided feedback on themes.

Results

In total, 55 interviews were conducted with 23 EIP practitioners and managers, 13 EIP service users, 10 carers, eight GPs (salaried or partners), and two commissioners. One interview was dyadic (a joint interview with a service user and carer). All EIP practitioners and managers were recruited through MHTs. Eleven service users were recruited through MHTs and two through service user groups. Two carers were recruited through an MHT and eight through social media and online networks. Commissioners and GPs were recruited through professional networks. Twenty-six people replied to our social media service user and carer recruitment flyer but either did not reply to follow-up contacts, or, following a screening call conducted by the first author, were found to be ineligible as they had not

been under EIP care, or were thought to be 'imposter participants' claiming to be EIP service users or carers but unable to answer questions about EIP care. 18 Interviews lasted between 25 and 50 minutes.

Key participant characteristics are presented in Tables 1–4. Most service users disclosed physical and/or mental health conditions in addition to the diagnosis of psychosis. A minority of carers (n = 3) reported physical and/or mental health diagnoses. Five carers lived with the person they provided care for.

Our findings are presented under four themes: barriers to accessing EIP services; perceived value of EIP services; lost connection with primary care; and discharge planning. Illustrative data are presented to support the analysis along with identifiers.

Barriers to accessing EIP services

EIP practitioners felt that the initial referral process was straightforward:

'So, anyone pretty much can refer directly into EIP, so GPs can come directly even though we have what's called a Single Point of Access [SPA] within the trust, so all GPs usually refer into that SPA service we call it ... So if there's a family member worried about their loved one ... and people themselves can refer ... so pretty

much open-door referral policy, yes.' (Healthcare practitioner [HCP]16)

However, GPs described limited knowledge of the criteria for admission to EIP care. Many GPs expressed frustration, a sense of powerlessness at rejected referrals, and concern for patients' safety and wellbeing:

'I referred her to the early intervention, and they did an assessment and basically felt that she wasn't ill enough to be under their care and just to refer her to the community mental health team because these were just symptoms of anxiety and depression and stress, which I felt was not the route that I would've suggested ... So it was quite upsetting ... I felt really disappointed for this lady because it was a big thing for her to disclose it to me ... I was saying, "Well there's this service called Early Intervention and I think this would be really good", almost trying to win her around because she was embarrassed to talk about it and then had to come back and be like, "Yeah, and the referral was rejected", which obviously doesn't fill her with confidence.' (GP1)

Perceived value of EIP services

EIP staff described the breadth of care offered, and the holistic approach taken, with an emphasis on patient-centred engagement and relationship building being key to the success of the EIP model:

| Table 1. Service user demographics | | | | | |
|------------------------------------|--------|------------|------------------------------|---------------------------------|-------------------------------|
| Participant ID | Sex | Age, years | Ethnic background | Living circumstances | Employment |
| SU1 | Female | 32 | White British | Lives with partner and children | Stay-at-home parent |
| SU2 | Female | 28 | White British | Lives alone | Unemployed |
| SU3 | Female | 28 | White and Black Caribbean | Lives with partner and child | Part-time employed/student |
| SU4 | Female | 26 | Mixed/multiple ethnicity | Lives with father | Unemployed |
| SU5 | Female | 43 | White Polish | Lives with partner and child | Full-time employed |
| SU6 | Female | 28 | White British | Lives with father | Unemployed |
| SU7 | Female | 61 | White British | Lives with partner | Unemployed |
| SU8 | Female | 57 | Black Caribbean | Lives alone | Unemployed |
| SU9 | Male | 43 | Black British | Lives with partner | Full-time employed |
| SU10 | Female | 64 | White British | Lives with partner | Unemployed |
| SU11 | Male | 25 | White British | Lives alone | Full-time employed |
| SU12 | Male | 43 | White British | Lives with partner and children | Full-time employed |
| SU13 | Male | 45 | White British | Lives alone | Unemployed |

| - 11 - | | | |
|-------------|----------|---------|--------|
| Table 2. | Carer d | lemogra | innics |
| I dibite L. | our or o | | Pilico |

| | | Relationship to person | | | |
|-------------------|--------|---------------------------|------------|----------------------|--------------------|
| Participant ID | Sex | with psychosis | Age, years | Ethnic background | Employment |
| CAR1 | Male | Brother | 25 | Asian | Full-time employed |
| CAR2 | Male | Friend | 26 | Black British | Part-time employed |
| CAR3 | Female | Niece | 25 | Indian British | Part-time employed |
| CAR4 | Female | Mother | 65 | White British | Retired |
| CAR5 | Male | Husband | 59 | White British | Full-time employed |
| CAR6 | Male | Grandchild | 23 | Black British | Part-time employed |
| CAR7 | Female | Mother | 59 | White British | Part-time employed |
| CAR8 | Male | Partner | 65 | White British | Retired |
| CAR9 | Female | Former partner | 45 | Bangladeshi | Full-time employed |
| CAR10 | Female | Mother | 45 | White British | Part-time employed |

| Table 3. GP background | | | | |
|---|--------|---------------------|------------------|------------|
| Participant ID | Sex | Years as GP | Size of practice | Geography |
| GP1 | Female | 10 | 25k | Inner city |
| GP2 | Male | 5 | 26k | Urban |
| GP3 | Male | 5 years as partner | 20k | Rural |
| GP4 | Female | 25 | N/Aª | N/Aª |
| GP5 | Male | 11 years as partner | 14k | Suburban |
| GP6 | Female | 19 | 14k | Urban |
| GP7 | Female | 13 | 10k | Semi-rural |
| GP8 | Female | 4 | 3k | Urban |
| °GP4 worked as a locum across several practices in different localities. N/A = not available. | | | | |

'We have the ability to concentrate on engagement and ... proper patient-centred care, it's not just a throwaway line, and we have access to vocational support, to psychology, to social recovery support.' (HCP15)

Service users talked about being supported and understood by their care coordinator:

'[Name of care coordinator] knew so much about me as well, so after three years I didn't have to tell her who somebody was or anything because she just knew and you'd be like "How's this person?", "How's that person?", she knew everything about me, so I had no worries there.' (Service user [SU]2)

EIP care coordinators were recognised as essential to the success of EIP relationship building:

'The relationship and the attachment they have to us is that kind of key bedrock of what helps people.' (HCP17)

Service users described how the EIP service had given them better understanding of their condition, which helped them feel less isolated and more in control:

'It's talking to people, people who have really good expertise in psychosis. Because I had no idea what it was. I had all these awesome people around us, like the CPN [community psychiatric nurse] and the support worker. I didn't feel alone, like I could go to them.' (SU6)

GPs had limited awareness of the duration of EIP care, what EIP care involves, and the discharge process:

'I saw a lady the other day and she was actually discharged from them because she'd been under them for three years, which apparently is the maximum amount of time that they can be under them, which I wasn't aware of.' (GP1)

Some GPs were also unsure about the respective roles of EIP and crisis teams:

'I think we'd send them to the crisis team. I mean, I never heard the term EI [Early Intervention] until recently.' (GP6)

Therefore, while supportive relationships were reported to be established between service users and EIP practitioners during the period of EIP care, GPs reported little awareness of the function of EIP services.

Lost connection with primary care

Service users appreciated the strong supportive relationships they had established with EIP practitioners but reported little to no contact with their GP. As a consequence, service users preferred to speak with EIP practitioners about their physical and mental health:

'The GP practice don't know an awful lot really, I think it's good that the hospital [EIP service] keeps you there quite a few years, I think it's nice really, you don't have to deal with your normal GP. Because like I said, I don't actually know who my GP is.' (SU2)

GPs suggested that their role was to refer patients into EIP services (directly or through a crisis team); they would then have little involvement while patients were under EIP care. GPs reported little engagement with EIP care coordinators during the period of EIP care or planning for discharge:

'I think the experiences I have had, have given me a reasonable level of trust that those initial stages are handled pretty well in a way that I don't have to be overly involved as a GP you know, beyond the referral stage.' (GP3)

Physical health checks and monitoring for service users were reported to be undertaken by EIP teams:

'So, we're the people meant to be responsible for that. So, the cardiometabolic risk factors, so bloods and BMI [body mass index] and blood pressure and smoking, et cetera, and particularly regarding the prescribing of antipsychotics. So, we're responsible for monitoring that. We don't ask the GPs to do it.' (HCP10)

GPs were perceived by EIP practitioners and carers to have responsibility for physical health checks only when EIP services lacked capacity:

'Unfortunately, we don't have a physical health team attached to us at the moment. So, it would be sharing that with the GP and what have you.' (HCP22)

Table 4. Early Intervention in Psychosis practitioner, manager, and commissioner background

| Participant ID | Sex | Job title |
|-------------------|--------|--|
| HCP1 | Female | Clinical lead, EIP |
| HCP2 | Female | EIP team manager |
| НСР3 | Female | EIP team manager |
| HCP4 | Female | Care coordinator/mental health nurse |
| HCP5 | Male | Consultant psychiatrist |
| НСР6 | Male | Advanced clinical practitioner/social worker |
| НСР7 | Female | Care coordinator/mental health nurse |
| НСР8 | Female | Community psychiatric nurse |
| НСР9 | Male | Consultant psychiatrist |
| HCP10 | Male | Consultant psychiatrist |
| HCP11 | Female | EIP team manager |
| HCP12 | Male | Psychologist |
| HCP13 | Male | Case manager |
| HCP14 | Male | EIP team manager |
| HCP15 | Female | Clinical nurse specialist |
| HCP16 | Female | EIP team manager |
| HCP17 | Male | Principal psychologist |
| HCP18 | Female | Principal psychologist |
| HCP19 | Male | Community psychiatric nurse |
| HCP20 | Female | Clinical lead, EIP |
| HCP21 | Female | Care coordinator/mental health nurse |
| HCP22 | Male | Community psychiatric nurse |
| HCP23 | Female | Peer support worker |
| COM1 | Male | Commissioner/senior programme manager |
| COM2 | Male | Commissioner/senior programme delivery lead |
| 510 5 1 1 1 | | |

EIP = Early Intervention in Psychosis.

'I think for that they referred him to his GP because I think they were overloaded with people wanting physical help ... and he has been using the same GP for many, many years, so sent him in for the GP and then the GP just conducted physical health evaluations.' (Carer [CAR] 3)

One GP, working in a small practice, reported that they continued to conduct physical health checks with EIP service users as a way of 'touching base' and maintaining a relationship:

'We still try and contact the patient and we would still try and have a review in house. That review might be that we say, "Look, we'd like to see you ... would you mind coming in just to meet us so we can just touch base with you, see how you're doing and whether there's anything else we need to help you with?"' (GP8)

In addition, one service user did have an established relationship with the GP, which continued while he was under EIP care.

Discharge planning

EIP practitioners and managers reported that discharge from EIP was planned well in advance. This was as a result of a recovery-centred approach and long waiting lists for CMHTs:

'So, we start thinking about it sometimes six or twelve months ahead and we have a chat about what a person's needs might be.' (HCP10)

'I think we work towards discharge from very early on. Although it's a three-year pathway, we are working ... this is our ideal, so we're working towards discharge. I don't think we get to the end of the three years and go, "Oh, we need to do some discharge planning." (HCP14)

For discharge to the CMHT, it was reported that handover meetings were held involving the EIP care coordinator, new CMHT care coordinator, and service user. EIP professionals were aware that this transition could be difficult and required a personalised approach:

'So, we would generally try and organise a couple of meetings joint with the case manager from the other team. So, yeah, just sit there and talk, talk about what we've done, what the person wants to do, and just so they feel comfortable with their new case manager before we fully discharge them. And we can stretch that out a bit as well if we think someone is particularly anxious about the transition.' (HCP13)

Discharge planning to primary care was seen by EIP practitioners as more straightforward, with little contact needed other than via a discharge letter:

'I think the liaison is maybe a little less comprehensive than with the adult mental health teams, just because of the nature of the sort of care that's expected afterwards. So, we would send the GP a letter, to update the GP about the person. There wouldn't necessarily be a meeting or a discussion; it would be information sharing by letter, really.' (HCP11)

'If we don't feel that they need to be in secondary care, we'll just sort of like discharge them back to their GP ... obviously we can just write to the GP and discharge the care back and the consultant will send their letter straight to the GP as well.' (HCP7)

EIP practitioners felt that current communication with GPs was sufficient:

'We send a letter usually, but if the GP needed to be involved [in meetings] we'd invite the GP. It's very rare GPs come, we do invite them to meetings and professional meetings sometimes talking about somebody's care, but yeah it would be a letter. And the GP could always contact us.' (HCP16)

GPs, however, expressed concerns about EIP discharge communication being limited to letters, which could be received some time after discharge:

'Well, there have been instances where I haven't been aware the person has been discharged. And I can't remember ever being in a situation where somebody's deliberately made a point to phone me to say, "your patient's being discharged". And

often there's a delay in paperwork so that's my impression of it. So, I don't think there is a clear communication to the GP at point of discharge.' (GP5)

Detailed discharge letters can be an important therapeutic tool for service users and provide a shared narrative of their journey through EIP. GPs, however, reported them to be an unwieldy form of communication:

'The mental health discharge letters they can be seven, eight, ten pages long telling us great depth about the full mental state, exam that they've done, and things like that. And then just very hard within there to see what the kind of salient points of the case are. You know, the detail is there, but I'm run off my feet in the clinic and I've been given this one sort of eight-page letter of someone who's been in the services a while and they've now felt is fit to be discharged.' (GP2)

GPs suggested that service users discharged to primary care could still have substantial ongoing support needs. They expressed concern about managing complex antipsychotic medication without specialist support:

'The main challenges that I have had have been when the mental health services have wanted to discharge people on, like, say new antipsychotic medications, either medication or depot [injection], but you know, really that should be all under kind of formal shared care agreements between secondary care and primary care, and there's an increasing push for people to be discharged from the team I think and GPs to take full ownership of the antipsychotic medications.' (GP3)

One GP suggested that a transitional approach to discharge could improve communication and joint working between primary care and EIP care, but acknowledged this would require additional time:

'Maybe [service users could] stay under the early intervention team for longer, but more on an arm's distance. And so, as GPs, we could kind of see them regularly, but then if we felt we had concerns we could liaise with the early intervention and try and help manage them more remotely, but with their experience, their quidance ... I think there would definitely be scope for us to work more with the early intervention team, but it's just finding the time really.' (GP1)

Service users and carers expressed worry and uncertainty about discharge from EIP:

'I guess I am a bit afraid about what comes next because it does kind of feel like somebody has took the lead off, kind of thing. You won't be able to talk to the people like you've formed a connection with for three years and that's kind of scary. But you know, I feel like I have the tools to deal with it than I did three years ago.' (SU6)

'Personally, I'm terrified [of discharge] because all my help, I know where to go, I know where to get the help immediately if I need it. Any problems I know what to do and I know his care coordinator so well that I feel comfortable.' (CAR7)

One service user described how she had relapsed after discharge to primary care, and felt that her concerns about discharge had contributed to this:

'I was really sad to be honest, it's like a weird feeling, it's been so long just having someone there as like a bit of a comfort blanket and then all of a sudden they're not going to be there any more and you just think, "Oh no." Because anything I needed I could call her or text her whenever I wanted. So obviously not having that there was pretty worrying." (SU2)

Most service users described concern about not having continuity with one GP, having to re-tell their story following discharge:

'What I have asked is, although I've got a named GP at my doctor's, it has changed where it can be multiple people that you've not really seen before. So, I did ask my consultant yesterday if I could ... I think for me, it's about someone who just has a little bit of an idea about what's gone on previously ... but it's a bit annoying, I think, especially when you are not feeling good mentally, when you're having to repeat yourself and tell your story time and time again.' (SU1)

Carers also felt disconnected from primary care, expressing fear they would not be able to access support for their loved ones after discharge:

'My first port of call would be the GP. I wouldn't know who I would be speaking to there ... and I would feel like we have to go back to square one again and start again. And so I would be doing that out of desperation with no confidence that there would be an appropriate reaction.' (CAR8)

Although complex medication regimens were of concern to GPs, medication management could offer a mechanism for service users to build a relationship with their GP:

'One thing that was really helpful and has now stopped is she was having her depot at the GP and she was absolutely happy to *go.* She felt comfortable with the [practice] nurses, and they had a chat. It wasn't the same one I don't think, and she'd say, "Oh, it was [Name] today and we talked about this." So that did build a certain relationship with the practice.' (CAR4)

Discussion

Summary

This qualitative study illustrates the tensions across the primary and specialist care interface for EIP services in England. GPs and service users reported difficulties in accessing EIP services and challenges at the point of discharge. As referrals to EIP services are often made at crisis point, service users may not have been known to their GP before referral, and there may not be a pre-existing relationship to pick up on discharge from the service. We report that service users under an EIP service may have little contact with a GP. Further, physical health monitoring may not be done in primary care.

Experiences while in EIP services were described as positive by most service users and carers in this study, with service users highlighting the benefits of regular and continuing support from care coordinators. Discharge to CMHTs was reported to be well planned and proactive, although sometimes impeded by long waiting times for admission to CMHTs. This contrasted with discharge to primary care being seen by EIP professionals as needing no more than a letter to the GP informing them of the discharge. Both service users and carers described feelings of abandonment when discharged to general practice, often to a GP they did not know.

Strengths and limitations

This study explores experiences of EIP services from multiple perspectives and illustrates the complexity of the patient journey through these services. We recruited a diverse sample and analysis was conducted by a multidisciplinary

There are limitations to this study. Of the 13 service user participants, only four were male, yet most people in EIP services are male. 19 Seven of the ten carer participants were caring for male service users, which gave additional insight. Four service users were non-White despite attempts to recruit an ethnically diverse

Funding

This study/project (EXTEND: Personalised Care for Early Psychosis: https://www.psych.ox.ac.uk/ research/extend) is funded by the National Institute for Health and Care Research (NIHR) under its Programme Grants for Applied Research (reference: NIHR203277). The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care. Carolyn A Chew-Graham is part-funded by the NIHR Applied Research Collaboration West Midlands.

Ethical approval

Research ethics approval was obtained in September 2022 from the North of Scotland Research Ethics Committee (reference: 22/NS/0113) and Health Research Authority (Integrated Research Application System ID: 313927).

Provenance

Freely submitted, externally peer reviewed.

Competing interests

David Shiers is an expert advisor to the National Institute for Health and Care Excellence (NICE) centre for guidelines; views expressed are the authors and not those of NICE. The other authors have declared no competing interests.

Acknowledgements

We would like to thank all interview participants for their time. Thank you to members of the service user and carer Involvement Group (EXTEND-InG: https://www.psych.ox.ac.uk/research/extend/extend-ing) for their contribution to the EXTEND qualitative study, particularly their input to data analysis.

Open access

This article is Open Access: CC BY 4.0 licence (http://creativecommons.org/licences/by/4.0/).

Discuss this article: bjgp.org/letters

sample through MHTs and social media. Service user participants were all stable in terms of their mental health at the point of interview. Carers shared experiences of service users who were unwell, to support a broader understanding. Most service user participants had become unwell before the COVID-19 pandemic; more recent experiences of a new psychotic illness may reveal important differences. We gave service users and carers the opportunity to be interviewed together if they preferred but only one dyadic interview was completed.

Comparison with existing literature

Our study adds to the very limited research on patients' and carers' experience of discharge to primary care from EIP services, even though this is the most common outcome following EIP service input.^{6,7} Service users value the continuity of care provided by EIP services — particularly the opportunity to build a therapeutic relationship with a named care coordinator — which is consistent with previous literature.20 One study has shown that continuity of care in EIP services results in higher service user satisfaction and better health outcomes.5 Similarly, continuity for people with severe mental illness (SMI) leads to better health outcomes, reduced hospital admission, and is more cost-effective.21

Supporting the physical health needs of people with SMI is one of the five clinical priorities of the CORE20PLUS5 approach,²² and is particularly urgent given the 15-year mortality gap between people with SMI and the general population.²³ The Positive Cardiometabolic Health Resource (Lester update) emphasised the importance of physical health monitoring from the earliest stages of a psychotic illness, particularly when antipsychotic medication is prescribed,¹¹ as the risk of weight gain in the first 12 months requires early intervention.⁹

The lack of contact with primary care while people are managed by EIP services reduces opportunities for service users and carers to establish relationships with general practice and engage with physical health monitoring.²⁴

Implications for practice

We suggest it would be beneficial for closer connections to be maintained between patients in EIP services and their registered general practice — there are various ways to achieve this. This could

be a 'named GP' to take over prescribing on discharge and/or a healthcare assistant or practice nurse tasked with conducting the physical health check.25 Mental health practitioners, embedded in primary care teams (as part of the Additional Roles Reimbursement Scheme),²⁶ could maintain contact with service users under EIP care and liaise with care coordinators, GPs, and carers to plan discharge. A joint consultation between the service user, their carer (if they wish), care coordinators, and primary care clinicians would support effective transitions. Supporting the service user and carer to understand when and how to access primary care would be valuable, particularly if the person were to become unwell again. Closer connections with primary care would also promote relationship-based care, which is already a hallmark of the care that people receive in EIP services. Our findings could be used to support the education and training of primary and specialist care practitioners, and commissioners.

We suggest that current discharge letters are not used effectively to communicate with primary care. A standard structured discharge letter should be jointly developed, which summarises the key points of importance for primary care clinicians.²⁷

Primary care also has a role in supporting carers,³ so maintaining contact with, and offering support to, the carer of a person being managed by EIP services is vital.

References

- Marshall M, Lewis S, Lockwood A, et al.
 Association between duration of untreated psychosis and outcome in cohorts of first-episode patients: a systematic review. Arch Gen Psychiatry 2005; 62(9): 975–983.
- Correll CU, Galling B, Pawar A, et al.
 Comparison of early intervention services vs treatment as usual for early-phase psychosis: a systematic review, meta-analysis, and meta-regression. JAMA Psychiatry 2018; 75(6): 555–565.
- Onwumere J, Shiers D, Chew-Graham C. Understanding the needs of carers of people with psychosis in primary care. Br J Gen Pract 2016; DOI: https://doi.org/10.3399/ bjgp16X686209.
- Ahmed S, Peters KZ, Chakraborty N.
 Discharges from an early intervention in psychosis service: the effect of patient characteristics on discharge destination.

 Early Interv Psychiatry 2019; 13(4): 740–744.
- Loughlin M, Berry K, Brooks J, Bucci S. Moving on from early intervention for psychosis services: service user perspectives on the facilitators and barriers of transition.

- Early Interv Psychiatry 2019; 13(6): 1396-1403.
- 6. Lester H, Khan N, Jones P, et al. Service users' views of moving on from early intervention services for psychosis: a longitudinal qualitative study in primary care. Br | Gen Pract 2012; DOI: https://doi. org/10.3399/bjgp12X630070.
- 7. Puntis S, Pappa S, Lennox B. What happens after early intervention? Mapping early intervention in psychosis care pathways in the 12 months after discharge. Early Interv Psychiatry 2024; 18(1): 49-57.
- 8. Woodward S, Bucci S, Edge D, Berry K. Barriers and facilitators to 'moving on' from early intervention in psychosis services. Early Interv Psychiatry 2019; 13(4): 914-921.
- Alvarez-Jiménez M, González-Blanch C, Crespo-Facorro B, et al. Antipsychoticinduced weight gain in chronic and firstepisode psychotic disorders: a systematic critical reappraisal. CNS Drugs 2008; 22(7): 547-562.
- 10. Nielsen RE, Uggerby AS, Jensen SOW, McGrath JJ. Increasing mortality gap for patients diagnosed with schizophrenia over the last three decades: a Danish nationwide study from 1980 to 2010. Schizophr Res 2013; 146(1-3): 22-27.
- 11. Perry BI, Holt RIG, Chew-Graham CA, et al. 2023 update (with acknowledgement to the late Helen Lester for her contribution to the original 2012 version) Positive Cardiometabolic Health Resource: an intervention framework for people experiencing psychosis and schizophrenia. 2023 update. 2023. https://www.rcpsych. ac.uk/docs/default-source/improving-care/ ccqi/national-clinical-audits/ncap-library/ eip-2024/ncap-lester-tool-interventionframework.pdf (accessed 16 Jul 2024).

- 12. Firth J, Siddiqi N, Koyanagi A, et al. The Lancet Psychiatry Commission: a blueprint for protecting physical health in people with mental illness. Lancet Psychiatry 2019; 6(8): 675-712.
- 13. Braun V, Clarke V. Using thematic analysis in psychology. Qual Res Psychol 2006; 3(2): 77-101.
- 14. Creswell JW. Data analysis and representation. In: Creswell JW, ed. Qualitative inquiry and research design: choosing among five approaches. 2nd edn. Thousand Oaks, CA: Sage Publications, 2007: 147-176.
- 15. Malterud K, Siersma VD, Guassora AD. Sample size in qualitative interview studies: guided by information power. Qual Health Res 2016; 26(13): 1753-1760.
- 16. Nowell LS, Norris JM, White DE, Moules NJ. Thematic analysis: striving to meet the trustworthiness criteria. Int J Qual Methods 2017; DOI: 10.1177/1609406917733847.
- 17. Gale NK, Heath G, Cameron E, et al. Using the framework method for the analysis of qualitative data in multi-disciplinary health research. BMC Med Res Methodol 2013; 13:
- 18. Ridge D, Bullock L, Causer H, et al. 'Imposter participants' in online qualitative research: a new and increasing threat to data integrity? Health Expect 2023; 26(3): 941-944.
- 19. Royal College of Psychiatrists. National appendices for England: Early Intervention in Psychosis audit 2021/22. 2022. https:// www.rcpsych.ac.uk/docs/default-source/ improving-care/ccqi/national-clinical-audits/ ncap-library/eip-2021-22/ref-333---ncapeip-england-appendices-20220713.pdf (accessed 23 Jul 2024).

- 20. Lester H, Birchwood M, Bryan S, et al. Development and implementation of early intervention services for young people with psychosis: case study. Br J Psychiatry 2009; 194(5): 446-450.
- 21. Adair CE, McDougall GM, Mitton CR, et al. Continuity of care and health outcomes among persons with severe mental illness. Psychiatr Serv 2005; 56(9): 1061-1069.
- 22. NHS England. Core20PLUS5 (adults) an approach to reducing healthcare inequalities. https://www.england.nhs.uk/ about/equality/equality-hub/nationalhealthcare-inequalities-improvementprogramme/core20plus5 (accessed 23 Jul
- 23. Public Health England. Severe mental illness (SMI) and physical health inequalities: briefing. 2018. www.gov.uk/government/ publications/severe-mental-illness-smiphysical-health-inequalities/severe-mentalillness-and-physical-health-inequalitiesbriefing (accessed 16 Jul 2024).
- 24. Jones R, Major B, Fear C. Schizophrenia in a primary care setting. Curr Psychiatry Rep 2015; 17(10): 84.
- 25. NHS England. Quality and Outcomes Framework guidance for 2023/2024. 2024. https://www.england.nhs.uk/publication/ quality-and-outcomes-framework-guidancefor-2023-24 (accessed 16 Jul 2024).
- 26. NHS England. Expanding our workforce. https://www.england.nhs.uk/gp/expandingour-workforce (accessed 16 Jul 2024).
- 27. Nilforooshan R, Weston L, Sachdeva D, et al. What information do general practitioners expect in letters from mental health services? London J Prim Care 2009; 2(1): 43-45